Treatment of anorexia nervosa: a multimethod investigation translating experimental neuroscience into clinical practice

Ulrike Schmidt, 1,2*† Helen Sharpe, 1† Savani Bartholdy, 1 Eva-Maria Bonin, 3 Helen Davies, 1 Abigail Easter, 1,4 Elizabeth Goddard, 1 Rebecca Hibbs, 1 Jennifer House, 1 Alexandra Keyes, 1 Pooky Knightsmith, 1 Antonia Koskina, 1 Nicholas Magill, 1 Jessica McClelland, 1 Nadia Micali, 1,4 Simone Raenker, 1 Bethany Renwick, 1 Charlotte Rhind, 1 Mima Simic, 2 Lot Sternheim, 1 Sabine Woerwag-Mehta, 1 Jennifer Beecham, 3,5 lain C Campbell, 1 Ivan Eisler, 2,6 Sabine Landau, 7 Susan Ringwood, 8 Helen Startup, 2 Kate Tchanturia 1,2 and Janet Treasure 1,2

¹Department of Psychological Medicine, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

²South London and Maudsley NHS Foundation Trust, London, UK

³Personal Social Services Research Unit, London School of Economics and Political Science, London, UK

⁴Institute of Child Health, University College London, London, UK

⁵Personal Social Services Research Unit, University of Kent, Canterbury, UK

⁶Department of Child Psychiatry, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

⁷Department of Biostatistics, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

⁸Beat (Eating Disorders Association)

^{*}Corresponding author ulrike.schmidt@kcl.ac.uk

[†]These authors contributed equally to the report and are joint first authors

Declared competing interests of authors: Ulrike Schmidt, Sabine Landau and Janet Treasure received salary support from the South London and Maudsley NHS Foundation Trust's Mental Health Biomedical Research Centre. Savani Bartholdy reports grants from the National Institute for Health Research (NIHR) Biomedical Research Council outside the submitted work. Charlotte Rhind reports grants from the Psychiatry Research Trust during the conduct of the study. Janet Treasure reports personal fees from Routledge (publishers) outside the submitted work. Rebecca Hibbs reports grants from the NIHR Research for Patient Benefit Programme (PB-PG-0609-19025) during the conduct of the study.

Published August 2017 DOI: 10.3310/pgfar05160

Plain English summary

Treatment of anorexia nervosa

Programme Grants for Applied Research 2017; Vol. 5: No. 16 DOI: 10.3310/pgfar05160

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

A norexia nervosa (AN) is an eating disorder (ED) with serious consequences for sufferers and their families. We do not currently know how best to treat people with AN, especially those with a severe illness. We carried out 10 studies to improve our understanding.

We evaluated six interventions: teacher training on managing EDs, school lessons aiming to prevent EDs, an outpatient talking therapy for adults, an inpatient treatment for adults, an intervention for carers of people with AN, and an online intervention to prevent relapse after treatment.

We also studied two things that will help us to develop better treatments in the future: the levels of physical activity (PA) in people with AN, and fertility problems in women with EDs and the growth and diet of their children.

Finally, we studied the importance of specialist services in treating EDs and the cost of treatments for AN and the total costs of AN in England each year.

We found that participants liked the new interventions; most of the interventions were more helpful than treatment as usual; the new outpatient treatment was as helpful as an established gold-standard treatment and better liked; people with EDs were different from healthy individuals in some aspects of their PA, their fertility, and the growth and diet of their children; having access to specialist services meant that patients had more positive experiences; and AN costs between £45M and £230M in England each year.

Overall, these studies have greatly improved our understanding of how best to help individuals with AN.

Programme Grants for Applied Research

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Programme Grants for Applied Research programme

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was set up in 2006 to produce independent research findings that will have practical application for the benefit of patients and the NHS in the relatively near future. The Programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director.

The programme is a national response mode funding scheme that aims to provide evidence to improve health outcomes in England through promotion of health, prevention of ill health, and optimal disease management (including safety and quality), with particular emphasis on conditions causing significant disease burden.

For more information about the PGfAR programme please visit the website: http://www.nihr.ac.uk/funding/programme-grants-for-applied-research.htm

This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-0606-1043. The contractual start date was in August 2007. The final report began editorial review in February 2016 and was accepted for publication in October 2016. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2017. This work was produced by Schmidt et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Programme Grants for Applied Research Editor-in-Chief

Professor Paul Little Professor of Primary Care Research, University of Southampton, UK

NIHR Journals Library Editor-in-Chief

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

Professor Matthias Beck Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk